Author's response to reviews

Title: Familial liability, obstetric complications and childhood development abnormalities in early onset schizophrenia: a case control study

Authors:

Francesco Margari (margari.f@psichiat.uniba.it)
Maria G Petruzzelli (giusypetruzzelli@libero.it)
Paola A Lecce (paolalecce@alice.it)
Orlando Todarello (ortod@psichiat.uniba.it)
Andrea De Giacomo (a.degiacomo@psichiat.uniba.it)
Eliana Lucarelli (elianalucarelli@libero.it)
Domenico Martinelli (domenicomartinelli@gmail.com)
Lucia Margari (l.margari@neurol.uniba.it)

Version: 3 Date: 14 March 2011

Author's response to reviews: see over
Object: Submission of revised manuscript

Dear Editor,

I send you the revised manuscript FAMILIAL LIABILITY, OBSTETRIC COMPLICATIONS AND CHILDHOOD DEVELOPMENT ABNORMALITIES IN EARLY ONSET SCHIZOPHRENIA: A CASE-CONTROL STUDY by Margari F. et al, for publication on your Journal as research article.

In response to the comments made by the reviewers we have improved the English and have checked the formatting of manuscript.

Looking forward to hearing from you, I send my kindest regards.

Sincerely,
Lucia Margari